Author's response to reviews

Title: Acceptability and implications of the rapid hepatitis C screening test among high-risk young people who inject drugs

Authors:

Benjamin Hayes (Benjamin.Hayes@ucsf.edu)
Alya Briceno (Abriceno@psg.ucsf.edu)
Alice Asher (Aasher@psg.ucsf.edu)
Michelle Yu (MYu@psg.ucsf.edu)
Jennifer Evans (Jevans@epi.ucsf.edu)
Judith A Hahn (Jhahn@epi-center.ucsf.edu)
Kimberly Page (Kpage@psg.ucsf.edu)

Version: 4 Date: 4 June 2014

Author's response to reviews: see over
June 2, 2014

Dear Dr. Zule,

We thank the Editors and Reviewers at BMC Public Health for the very positive response to our resubmitted manuscript “Preference, acceptability and implications of the rapid hepatitis C screening test among high-risk young people who inject drugs”. I have added a statement regarding ethical review and regarding the consent process (written) to the manuscript (Methods section, page 7). I hope that this revision meets the journal’s requirement and is suitable for publication. Please do not hesitate to contact me with any questions or clarifications.

Regards,

Kimberly Page

Kimberly Page, PhD, MPH, MS
Professor in Residence